

ORIGINAL

SBC Long Distance 5850 W. Las Positas Blvd. DISTRIBUTION CENTERSanton, CA 94588

03 JUL 18 AM 10: 12

July 16, 2003

Mr. Walter D'Haeseleer, Director Florida Public Service Commission 2540 Shumard Oak Boulevard Tallahassee, Florida 32399

ဆိ AH IO:

Advice Letter 78 of Southwestern Bell Communications Services, Inc. Re: d/h/a Southwestern Bell Long Distance d/b/a Nevada Bell Long Distance d/b/a Pacific Bell Long Distance d/b/a SBC Long Distance

Dear Mr. D'Haeseleer:

NUS

:OM

TR

ECR CL:

жС

MS ЖC

MH

Enclosed are an original and three (3) copies of revisions to Florida Tariff No. 2 of Southwestern Bell Communications Services, Inc. d/b/a Southwestern Bell Long Distance d/b/a Nevada Bell Long Distance d/b/a Pacific Bell Long Distance d/b/a SBC Long Distance ("SBCS, Inc."). These revisions have an issue date of July 16, 2003. SBCS, Inc. requests an effective date of July 22, 2003. The purpose of this tariff filing is to (1) add, revise, and delete definitions; (2) revise the rules and regulations governing term plan commitments; and (3) introduce the following new service offerings: Business Domestic Saver 15 Connections 3 Service, Business Long Distance 50 Connections 3 Service, Business Long Distance 100 Connections 3 Service, Business Domestic Saver 15 Connections 2 Service, Business Long Distance 50 Connections 2 Service, Business Long Distance 100 Connections 2 Service, Business Domestic Saver 15 Connections 1 Service, Business Long Distance 50 Connections 1 Service, Business Long Distance 100 Connections 1 Service, Business Domestic Saver 15 Connections 1 Plus Service 1 Year, Business Long Distance 50 Connections 1 Plus Service 1 Year, Business Long Distance 100 Connections 1 Plus Service 1 Year, Business Domestic Saver 15 Connections 1 Plus Service 2 Year, Business Long Distance 50 Connections 1 Plus Service 2 Year, Business Long Distance 100 Connections 1 Plus Service 2 Year, Business Domestic Saver 15 Connections 2 Plus Service, 1 Year, Business Long Distance 50 Connections 2 Plus Service 1 Year, Business Long Distance 100 Connections 2 Plus Service 1 Year, Business Domestic Saver 15 Connections 2 Plus Service 2 Year, Business Long Distance 50 Connections 2 Plus Service 2 Year, Business Long Distance 100 Connections 2 Plus Service 2 Year, Value Plans, and Business Unlimited Long Distance Plans.

DOCUMENT NUMPER-DATE

The following tariff sheets are being issued or revised:

68th Revised Sheet 2 22nd Revised Sheet 3 19th Revised Sheet 4 20th Revised Sheet 7 4th Revised Sheet 10.1.1 Original Sheet 10.1.2 Original Sheet 10.1.3 Original Sheet 10.1.4 5th Revised Sheet 12.0.1 Original Sheet 12.0.2 8th Revised Sheet 25 5th Revised Sheet 25.1 9th Revised Sheet 26 Original Sheet 26.1 2nd Revised Sheet 27.1 4th Revised Sheet 28 3rd Revised Sheet 29 7th Revised Sheet 29.2 7th Revised Sheet 30 11th Revised Sheet 32 6th Revised Sheet 33 4th Revised Sheet 33.1 5th Revised Sheet 34.1 1st Revised Sheet 35.1 4th Revised Sheet 93 4th Revised Sheet 97.1 4th Revised Sheet 97.2 6th Revised Sheet 97.3 4th Revised Sheet 97.4 4th Revised Sheet 97.5 12th Revised Sheet 98 **Original Sheet 98.1** 1st Revised Sheet 156.40 Original Sheet 156.40.1 Original Sheet 156.40.2 Original Sheet 156.40.3

Original Sheet 156.40.4 1st Revised Sheet 156.41 Original Sheet 156.41.1 Original Sheet 156.41.2 Original Sheet 156.41.3 Original Sheet 156.41.4 1st Revised Sheet 156.42 Original Sheet 156.42.1 Original Sheet 156.42.2 Original Sheet 156.42.3 Original Sheet 156.42.4 1st Revised Sheet 156.43 Original Sheet 156.43.1 Original Sheet 156.43.2 Original Sheet 156.43.3 Original Sheet 156.43.4 Original Sheet 156.43.5 1st Revised Sheet 156.44 Original Sheet 156.44.1 Original Sheet 156.44.2 Original Sheet 156.44.3 Original Sheet 156.44.4 1st Revised Sheet 156.45 Original Sheet 156.45.1 Original Sheet 156.45.2 Original Sheet 156.45.3 Original Sheet 156.45.4 1st Revised Sheet 156.46 Original Sheet 156.46.1 Original Sheet 156.46.2 Original Sheet 156.46.3 Original Sheet 156.46.4 Original Sheet 156.46.5 1st Revised Sheet 156.47 Original Sheet 156.47.1 Original Sheet 156.47.2

Original Sheet 156.47.3 Original Sheet 156.47.4 1st Revised Sheet 156.48 Original Sheet 156.48.1 Original Sheet 156.48.2 Original Sheet 156.48.3 Original Sheet 156.48.4 Original Sheet 156.57 Original Sheet 156.58 Original Sheet 156.59 Original Sheet 156.60 Original Sheet 156.61 Original Sheet 156.62 Original Sheet 156.63 Original Sheet 156.64 Original Sheet 156.65 Original Sheet 156.66 Original Sheet 156.67 Original Sheet 156.68 Original Sheet 156.69 Original Sheet 156.70 Original Sheet 156.71 Original Sheet 156.72 Original Sheet 156.73 Original Sheet 156.74 Original Sheet 156.75 Original Sheet 156.76 Original Sheet 156.77 Original Sheet 156.78 Original Sheet 156.79 Original Sheet 156.80 Original Sheet 156.81 Original Sheet 156.82 Original Sheet 156.83 Original Sheet 156.84 Original Sheet 156.85

Original Sheet 156.86	Original Sheet 156.107	Original Sheet 156.128
Original Sheet 156.87	Original Sheet 156.108	Original Sheet 156.129
Original Sheet 156.88	Original Sheet 156.109	1st Revised Sheet 184.8.2
Original Sheet 156.89	Original Sheet 156.110	1st Revised Sheet 184.8.3
Original Sheet 156.90	Original Sheet 156.111	1st Revised Sheet 184.8.4
Original Sheet 156.91	Original Sheet 156.112	1st Revised Sheet 184.8.5
Original Sheet 156.92	Original Sheet 156.113	1st Revised Sheet 184.8.6
Original Sheet 156.93	Original Sheet 156.114	1st Revised Sheet 184.8.7
Original Sheet 156.94	Original Sheet 156.115	1st Revised Sheet 184.8.8
Original Sheet 156.95	Original Sheet 156.116	1st Revised Sheet 184.8.9
Original Sheet 156.96	Original Sheet 156.117	1st Revised Sheet 184.8.10
Original Sheet 156.97	Original Sheet 156.118	Original Sheet 184.8.17
Original Sheet 156.98	Original Sheet 156.119	Original Sheet 184.8.18
Original Sheet 156.99	Original Sheet 156.120	Original Sheet 184.8.19
Original Sheet 156.100	Original Sheet 156.121	Original Sheet 184.8.20
Original Sheet 156.101	Original Sheet 156.122	Original Sheet 184.8.21
Original Sheet 156.102	Original Sheet 156.123	Original Sheet 184.8.22
Original Sheet 156.103	Original Sheet 156.124	Original Sheet 184.8.23
Original Sheet 156.104	Original Sheet 156.125	Original Sheet 184.8.24
Original Sheet 156.105	Original Sheet 156.126	Original Sheet 184.8.25
Original Sheet 156.106	Original Sheet 156.127	Original Sheet 184.8.26

So that our records will be complete, please date stamp the extra copy of the Advice Letter and return in the envelope provided. Any questions regarding this tariff filing should be referred to Barbara Lowe, Visiology, Inc., 16061 Carmel Bay Drive, Northport, Alabama 35475 who may be may be reached via telephone at (205) 330-1702.

Thank you for your assistance in this matter.

Yours truly,

Karen Bunkman

Karen Brinkman Project Administrator

Enclosures

CHECK SHEET

Sheets 1 through 191 of this Tariff are effective as of the date shown at the bottom of the sheet. Original and revised sheets as named below comprise all changes from the original Tariff.

SHEET	REVISION
1	2nd Revised Sheet
	68th Revised Sheet *
2 3	22nd Revised Sheet *
4	19th Revised Sheet *
5	5th Revised Sheet
6	11th Revised Sheet
7	20th Revised Sheet *
8	6th Revised Sheet
8.1	2nd Revised Sheet
8.2	1st Revised Sheet
8.3	7th Revised Sheet
9	9th Revised Sheet
9.1	28th Revised Sheet
9.2	2nd Revised Sheet
10	26th Revised Sheet
10.0.1	1st Revised Sheet
10.1	23rd Revised Sheet
10.1.0	Original Sheet
10.1.1	4th Revised Sheet *
10.1.2	Original Sheet *
10.1.3	Original Sheet *
10.1.4	Original Sheet *
10.2	9th Revised Sheet
10.3	Original Sheet
11	33rd Revised Sheet
12	28th Revised Sheet
12.0.1	5th Revised Sheet *
12.0.2	Original Sheet *
12.1	18th Revised Sheet
13	Original Sheet
14	Original Sheet
15	1st Revised Sheet
15.1	Original Sheet

* New or revised current Tariff filing.

SHEET	REVISION
16 17 18 19 20 21 22 23 24 25 25.1 25.2 26 26.1 27 27.1 28 29 29.1	6th Revised Sheet 2nd Revised Sheet 5th Revised Sheet 2nd Revised Sheet 2nd Revised Sheet 4th Revised Sheet 2nd Revised Sheet 2nd Revised Sheet 0riginal Sheet 8th Revised Sheet * 5th Revised Sheet * 0riginal Sheet 9th Revised Sheet * 5th Revised Sheet 2nd Revised Sheet 3rd Revised Sheet * 4th Revised Sheet * 4th Revised Sheet *
29.2	7th Revised Sheet *
29.2 30 31 31.1 32 33 33.1	7th Revised Sheet * 7th Revised Sheet * 4th Revised Sheet 1st Revised Sheet 11th Revised Sheet * 6th Revised Sheet * 4th Revised Sheet *

SHEET	REVISION
34	3rd Revised Sheet
34.1	5th Revised Sheet *
35	4th Revised Sheet
35.1	1st Revised Sheet *
36	Original Sheet
37	5th Revised Sheet
38	Original Sheet
39	1st Revised Sheet
40	Original Sheet
41	Original Sheet
42	1st Revised Sheet
43	3rd Revised Sheet
43.1	2nd Revised Sheet
43.2	1st Revised Sheet
44	2nd Revised Sheet
45	1st Revised Sheet
46	Original Sheet
47	Original Sheet
48	Original Sheet
49	Original Sheet
50	2nd Revised Sheet

SHEET	REVISION
85	3rd Revised Sheet
86	Original Sheet
87	Original Sheet
88	Original Sheet
89	Original Sheet
90	Original Sheet
91	1st Revised Sheet
92	1st Revised Sheet
93	4th Revised Sheet *
94	4th Revised Sheet
95	4th Revised Sheet
96	3rd Revised Sheet
97	1st Revised Sheet
97.1	4th Revised Sheet *
97.2	4th Revised Sheet *
97.3	6th Revised Sheet *
97.4	4th Revised Sheet *
97.5	4th Revised Sheet *
97.6	1st Revised Sheet
97.7	2nd Revised Sheet
97.8	1st Revised Sheet
98	12th Revised Sheet *
98.1	Original Sheet *
99	2nd Revised Sheet
100	2nd Revised Sheet
101	2nd Revised Sheet

* New or revised current Tariff filing.

SDC Long Distance	
	CHECK SHEET (continued)
SHEET	REVISION
156.20	3rd Revised Sheet
156.20.1	1st Revised Sheet
156.20.2	3rd Revised Sheet
156.21	2nd Revised Sheet
156.22	1st Revised Sheet
156.23	1st Revised Sheet
156.24	1st Revised Sheet
156.25	1st Revised Sheet
156.26	1st Revised Sheet
156.27	1st Revised Sheet
156.28	2nd Revised Sheet
156.29	1st Revised Sheet
156.30	1st Revised Sheet
156.31	1st Revised Sheet
156.32	1st Revised Sheet
156.33	1st Revised Sheet
156.34	1st Revised Sheet
156.35	2nd Revised Sheet
156.36	2nd Revised Sheet
156.37	2nd Revised Sheet
156.38	1st Revised Sheet
156.39	2nd Revised Sheet
156.40	1st Revised Sheet *
156.40.1	Original Sheet *
156.40.2	Original Sheet *
156.40.3	Original Sheet *
156.40.4	Original Sheet *
156.41	1st Revised Sheet *
156.41.1	Original Sheet *
156.41.2	Original Sheet *
156.41.3	Original Sheet *
156.41.4	Original Sheet *
156.42	1st Revised Sheet *
156.42.1	Original Sheet *
156.42.2	Original Sheet *
156.42.3	Original Sheet *
156.42.4	Original Sheet *

M - Material moved to Original Sheet 10.1.2

	CHECK SHEET (continued)	
SHEET	REVISION	
156.43	1st Revised Sheet *	М
156.43.1	Original Sheet *	
156.43.2	Original Sheet *	
156.43.3	Original Sheet *	
156.43.4	Original Sheet *	
156.43.5	Original Sheet *	
156.44	1st Revised Sheet *	М
156.44.1	Original Sheet *	
156.44.2	Original Sheet *	
156.44.3	Original Sheet *	
156.44.4	Original Sheet *	
156.45	1st Revised Sheet *	М
156.45.1	Original Sheet *	
156.45.2	Original Sheet *	
156.45.3	Original Sheet *	
156.45.4	Original Sheet *	
156.46	1st Revised Sheet *	М
156.46.1	Original Sheet *	
156.46.2	Original Sheet *	
156.46.3	Original Sheet *	
156.46.4	Original Sheet *	
156.46.5	Original Sheet *	
156.47	1st Revised Sheet *	Μ
156.47.1	Original Sheet *	
156.47.2	Original Sheet *	
156.47.3	Original Sheet *	
156.47.4	Original Sheet *	
156.48	1st Revised Sheet *	Μ
156.48.1	Original Sheet *	
156.48.2	Original Sheet *	
156.48.3	Original Sheet *	
156.48.4	Original Sheet *	
156.49	Original Sheet	Μ
156.50	Original Sheet	
156.51	Original Sheet	
156.52	Original Sheet	
156.53	Original Sheet	ļ
156.54	Original Sheet	ļ
156.55	Original Sheet	
156.56	Original Sheet	Μ

M - Material moved from 3rd Revised Sheet 10.1.1

Florida Tariff No. 2

	CHECK SHEET (continued)
SHEET	REVISION
SHEET	
156.57	Original Sheet *
156.58	Original Sheet *
156.59	Original Sheet *
156.60	Original Sheet *
156.61	Original Sheet *
156.62	Original Sheet *
156.63	Original Sheet *
156.64	Original Sheet *
156.65	Original Sheet *
156.66	Original Sheet *
156.67	Original Sheet *
156.68	Original Sheet *
156.69	Original Sheet *
156.70	Original Sheet *
156.71	Original Sheet *
156.72	Original Sheet *
156.73	Original Sheet *
156.74	Original Sheet *
156.75	Original Sheet *
156.76	Original Sheet *
156.77	Original Sheet *
156.78	Original Sheet *
156.79	Original Sheet *
156.80	Original Sheet *
156.81	Original Sheet *
156.82	Original Sheet *
156.83	Original Sheet *
156.84	Original Sheet *
156.85	Original Sheet *
156.86	Original Sheet *
156.87	Original Sheet *
156.88	Original Sheet *
156.89	Original Sheet *
156.90	Original Sheet *
156.91	Original Sheet *
156.92	Original Sheet *
156.93	Original Sheet *
156.94	Original Sheet *
156.95	Original Sheet *
156.96	Original Sheet *
156.97	Original Sheet *
156.98	Original Sheet *
156.99	Original Sheet *

SHEET

156.100

156.101

CHECK SHEET (continued)
REVISION
Original Sheet *

	0
156.102	Original Sheet *
156.103	Original Sheet *
156.104	Original Sheet *
156.105	Original Sheet *
156.106	Original Sheet *
156.107	Original Sheet *
156.108	Original Sheet *
156.109	Original Sheet *
156.110	Original Sheet *
156.111	Original Sheet *
156.112	Original Sheet *
156.113	Original Sheet *
156.114	Original Sheet *
156.115	Original Sheet *
156.116	Original Sheet *
156.117	Original Sheet *
156.118	Original Sheet *
156.119	Original Sheet *
156.120	Original Sheet *
156.121	Original Sheet *
156.122	Original Sheet *
156.123	Original Sheet *
156.124	Original Sheet *
156.125	Original Sheet *
156.126	Original Sheet *
156.127	Original Sheet *
156.128	Original Sheet *
156.129	Original Sheet *

SHEET	REVISION	
184.5	5th Revised Sheet	
184.6	1st Revised Sheet	
184.7	3rd Revised Sheet	
184.7.1	1st Revised Sheet	
184.8	3rd Revised Sheet	
184.8.1	1st Revised Sheet	
184.8.2	1st Revised Sheet *	
184.8.3	1st Revised Sheet *	
184.8.4	1st Revised Sheet *	
184.8.5	1st Revised Sheet *	
184.8.6	1st Revised Sheet *	
184.8.7	1st Revised Sheet *	
184.8.8	1st Revised Sheet *	
184.8.9	1st Revised Sheet *	
184.8.10	1st Revised Sheet *	
184.8.11	Original Sheet	
184.8.12	Original Sheet	
184.8.13	Original Sheet	
184.8.14	Original Sheet	
184.8.15	Original Sheet	
184.8.16	Original Sheet	
184.8.17	Original Sheet *	
184.8.18	Original Sheet *	
184.8.19	Original Sheet *	
184.8.20	Original Sheet *	
184.8.21	Original Sheet *	
184.8.22	Original Sheet *	
184.8.23	Original Sheet *	
184.8.24	Original Sheet *	
184.8.25	Original Sheet *	
184.8.26	Original Sheet *	

Μ

* New or revised current Tariff filing.

SHEET	REVISION	
185	5th Revised Sheet	Μ
186	4th Revised Sheet	
187	1st Revised Sheet	
188	1st Revised Sheet	
189	1st Revised Sheet	
189.1	6th Revised Sheet	
189.2	4th Revised Sheet	l
189.3	Original Sheet	1
189.4	1st Revised Sheet	
189.5	1st Revised Sheet	
189.6	Original Sheet	
189.7	1st Revised Sheet	
189.8	Original Sheet	М

* New or revised current Tariff filing.

Ν

N

Т

 \mathbf{T}

Ν

Ň

Ν

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS

30 Member Speed Calling: A service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.

Access Advantage PlusSM: A registered trademark of SBC Knowledge Ventures, Inc. Access Advantage Plus[®] is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.

Access or Access Line: A transmission line used to transmit voice and/or data calls from the Customer's Premises to a telephone company Serving Wire Center or a Company-designated POP or from a telephone company Serving Wire Center or a Company-designated POP to the Customer's Premises.

Affiliate: A company which has any of the following relationships with the Company; (1) directly or indirectly owns or controls it; (2) is directly or indirectly owned or controlled by it; or (3) is under common direct or indirect ownership with it.

Affiliated CLEC: A CLEC with which the Company has any of the following relationships: (1) owns or controls it; (2) is owned or controlled by it; or (3) is under common ownership with it.

Affiliated LEC: A LEC with which the Company has any of the following relationships: (1) owns or controls it; (2) is owned or controlled by it; or (3) is under common ownership with it.

Aggregation: The combining of a Customer's total usage across multiple BTNs into a group for the purpose of determining a common usage rate for call(s) associated with the individual BTNs.

Aggregation ID: Aggregation Identifier. Tags which BTNs are to be combined in a grouping.

Airline Mileage: The distance in mileage between two Serving Wire Centers whose position is specified by industry standards.

Alternate Routing PVC: Provides a logical connection to an alternate host site processor/server in the event of an outage at the primary location. Alternate Routing PVCs are utilized in the event of an outage at the primary location only, not typical day-to-day use.

Ameritech Centrex Service (ACS): A service provided by a SBC Affiliate.

Ancillary Charges: Charges for supplementary Services as set forth in this Tariff which may consist of both nonrecurring and monthly recurring charges.

ANI: Automatic Number Identification. A process used to identify the calling station. For example, Customers such as call centers pay for caller's telephone numbers to be sent to them simultaneously with their incoming toll free service calls.

ANSI: American National Standards Institute. A standards-setting, non-government organization, which develops and publishes standards for voluntary use in the United States.

Applicant: Any entity or individual who applies for Service under this Tariff.

Area of Service: The specific area(s) from which toll free calls will be allowed on a given TFS Number as decided by the Customer subscribing to that TFS Number.

ASR: Access Service Request. Used to request the provision of special access or Switched Access as specified in the tariff of the Local Access Provider.

ATM: Asynchronous Transfer Mode/Cell Relay Service. A high speed digital data Service utilizing cell-switching technology. Access speeds range form DS1 (1.544 Mbps) to OC12 (622 Mbps).

Authorization Level: An assigned level of calling privileges for VPN Service. Authorization Levels are assigned to private numbers when CSR features are utilized. If a caller encounters a call screening condition that restricts the caller from placing a particular call, the caller will be prompted to enter a VPN Authorization Code to override the restriction. If the Authorization Level assigned to the VPN Authorization Code is equal to or higher than the Authorization Level assigned to the private number, the call will be allowed to proceed. The caller is given three attempts to enter a valid VPN Authorization Code with the appropriate Authorization Level. If the Authorization Level is not sufficient, the call will be disconnected with an announcement after the third attempt.

Authorized User: A person, firm, corporation or other entity (including Customer) that 1) is authorized by the Customer to be connected to and utilize the Company's Services under the terms and regulations of this Tariff or 2) either is authorized by the Customer to act as the Customer in matters of ordering, changing or canceling Service or is placed in a position by the Customer, either through acts or omissions, to act as Customer in such matters. Such actions by an Authorized User shall be binding on Customer and shall subject Customer to any associated charges.

Auto RedialTM: A service provided by Southwestern Bell Telephone Company. Auto RedialTM is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.

B-Channel: The Bearer, or B, Channel. A 64 Kbps Channel that is a fundamental component of ISDN.

B8ZS (Bipolar 8 Zero Substitution): B8ZS is a DS1 line coding technique that enables digitized voice and data transmission at the rate of 1.536 Mbps independent of the number of consecutive zeros and pulse density requirements that are normally imposed on the T1 transmission line.

BAN: Billed Account Number.

Bandwidth: The total frequency band, in Hertz, allocated for a Channel.

Base Rate: The monthly recurring charge for Data Services without discounts.

BER: Bit error ratio. The percentage of received bits in error compared to the total number of bits received, expressed as a number the power of 10.

Issued: July 16, 2003

Effective: July 22, 2003

Т

| T

BTN: Billed Telephone Number. May consist of one or more WTNs.

Burst Rate: The upper Bandwidth limit the PVC is allowed to send data through the FRS network. The Burst Rate is limited by the actual physical Port access speed.

Business Customer: A Customer whose use of the Services is primarily or substantially for a business, professional, institutional, or occupational purpose.

Business Essentials SM : A service mark of SBC Knowledge Ventures, Inc. Business Essentials SM is a service provided by an SBC affiliate.	N
Business Preferred SM : A service mark of SBC Properties, L.P. Business Preferred SM is a service provided by a SBC affiliate.	
Business SolutionsSM: A service mark of SBC Properties, Inc. Business Solutions SM is a service provided by an SBC affiliate.	
Busy Call Forwarding: A service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	
Busy Call Forwarding-Extended: A service/feature associated with local exchange service which is defined in the tariff of a SBC Affiliate.	N
Call AgainTM: A trademark of Southern New England Telephone Company. Call Again TM is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Call BlockerTM: A trademark of Southwestern Bell Telephone Company and Southern New England Telephone Company. Call Blocker TM is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	T T
Call Blocking: A service provided by Southern New England Telephone Company. Call Blocking is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Call Forwarding: A service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Call in One: A service provided by Pacific Bell Telephone Company. Call In One is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Call Return: A service provided by Southwestern Bell Telephone Company. Call Return is a service/ feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N M
	 M

M - Material moved to Original Sheet 26.1

Effective: July 22, 2003

. -

	N N
	M M
Call Transfer Disconnect: A service or feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N
Call Waiting: A service or feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	
Call Waiting ID: A service or feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	 N
service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC	M/N N N
Plus is a service/feature associated with local exchange service which is defined in the tariff of the	M/N N N
CAP: Competitive Access Provider.	M
CARE: Customer Account Record Exchange.	
Carrier Common Line Charges: The charges the long distance companies pay to the local telephone companies for carrier common line access service which provides for the use of end user's telephone company provided common lines by subscribers for access to such end users to furnish interstate communications.	
Casual Caller: A caller that has not affirmatively selected the Company as its choice of a long distance service provider in advance of placing a long distance call.	 M

M - Material moved from 8th Revised Page 26

Issued: July 16, 2003

Ν

Ν

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS

Company: Southwestern Bell Communications Services, Inc. ("SBCS") d/b/a Southwestern Bell Long Distance ("SBLD"), d/b/a Nevada Bell Long Distance ("NBLD"), d/b/a Pacific Bell Long Distance ("PBLD"), and SBC Long Distance ("SBCLD").

Company-Provided: The switching, transmission, and other related telecommunications or computer equipment/facilities provided by the Company or by any combination of the Company, the LEC, or other authorized Third Party Vendors contracted by the Company.

CompleteLinkSM: A registered service mark of SBC Knowledge Ventures, Inc. CompleteLinkSM is a bundle associated with local exchange and long distance service that is defined in the tariff of an SBC Affiliate.

Corporate BAN: The highest level BAN on a hierarchy, is always an invoice BAN. VPN price plans will be attached at this level, which is the point of aggregation for the MAC and MMC.

CPE: Customer-Provided Equipment. Terminal equipment connected to the telephone network which is owned by the Customer or leased by the Customer from a supplier.

CPNIP: Customer Premises Network Interface Points.

Credit Card: Visa®, MasterCard®, or other Credit Cards issued by other companies the Company may accept.

CRC: Cycle Redundancy Check. A process used to check the integrity of a block of data.

CS: Controlled Slip: The occurrence at the receiving terminal of a replication or deletion of the information Bits in a frame.

CS Event: The occurrence of a Controlled Slip.

CSA: Canadian Standards Association. A non-profit, independent organization which operates a listing service for electrical and electronic materials and equipment.

CSR: Call Screen Routing.

CSU/DSU: Channel Service Unit/Data Service Unit.

Issued: July 16, 2003

Effective: July 22, 2003

N N

D

N N

Т

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS

Custom Biz SaverSM: A service mark of SBC Knowledge Ventures, Inc. Custom Biz SaverSM is a service provided by a SBC Affiliate.

Customer: A person or legal entity which subscribes to the Company's Services and thereby assumes responsibility for the payment of charges and compliance with the Company's Tariff.

Customer Commitment Date: The date in which the Company receives a firm commitment from a Customer for the provision of one of the Company's Data Service offerings.

Customer Premises/Customer's Premises: Location(s) designated by a Customer where Service is originated/terminated.

D-Channel: The Delta, Data, or D, Channel. Used to carry signaling and control information associated with B-Channels.

DACC: Directory Assistance Call Completion.

Data Services: Communication Services which are designed to allow the transfer of formatted information between points. Data Services include but are not limited to Private Line Service, Frame Relay Service, ATM Service, and Primary Rate ISDN (PRI).

DE: Discard Eligible. A Frame Relay service standard that specifies that data sent across a PVC in excess of that connection's CIR will be marked by the network as being eligible for discard by the network in the event of network congestion.

Dedicated Access: Where Customer's Premises has a non-switched connection to the POP selected by the Company for origination and or termination of calls. When the Dedicated Access is used for overlayed Switched Services, the Dedicated Access is referred to as a DVA line. When Dedicated Access is used to provide Data Services, the Dedicated Access is referred to as local loop.

Delayed Call Forwarding: A service or feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.

Direct-Dialed: A call placed by the caller without operator assistance (either live or automated).

Disaster Recovery PVCs: PVCs that allow for the implementation of logical connections between branch locations and a secondary processor/server center (disaster recovery site) should a non-recoverable disaster occur at the primary host site.

Diversity: Customer-designated routing which indicates a Customer-designated departure from the primary route, usually with physical separation.

DLCI: Data Link Connection Identifier. The address information assigned to Customer-designated end points used to identify PVCs and route frames of data.

DSO: Digital Signal level Zero. Composed of one 64 kiloBit Channel.

DS1: Digital Signal level One. Composed of twenty-four 64 Kbps Channels with a throughput capacity of 1.544 Mbps. Also called T-1.

Ν

Ν

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS

DS3: Digital Signal level Three. Composed of 28 DS1 Channels and operating at 44.736 Mbps. Also called T-3.

DSL: Digital Subscriber Line. A service provided by an SBC affiliate.

DVA: Dedicated Voice Access.

EABX: Electronic Automatic Branch e<u>X</u>change.

EFS: Error Free Seconds. A measure of the percentage of total seconds when measured over a consecutive thirty day period that do not contain Bit errors.

End User: The person or legal entity which uses the Service provided by the Company.

ES: Errored Second. A count of one-second intervals containing one or more CRC-6 code violations, or one or more CS events or one or more SEF events.

Equal Access: Enables the Customer to place long distance calls without the need to first dial a special code.

Exemption Certificate: A written notification provided by the Customer certifying that its dedicated facility should be exempted from the monthly Special Access Surcharge because (a) the facility terminates in a device not capable of interconnecting Service with the local exchange network or (b) the facility is associated with a Switched Access Service that is subject to Carrier Common Line Charges.

Extended Super-Frame Format: A DS1 framing standard. With this format twenty-four frames, instead of twelve, are grouped together.

F.C.C.: Federal Communications Commission or any succeeding agency.

FE: Framing-Bit Error or Framing Error. An error occurring when a receiver improperly interprets the set of bits within a Frame.

FE Event: The occurrence of a framing error or framing-Bit error.

Southwestern Bell Communications Services, Inc. d/b/a Southwestern Bell Long Distance d/b/a Nevada Bell Long Distance d/b/a Pacific Bell Long Distance d/b/a SBC Long Distance Florida Tariff No. 2 7th Revised Sheet 29.2 Cancels 6th Revised Sheet 29.2

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS Gbps: GigaBits per second. A billion Bits per second.

Group A Large Package: Features associated with the provision of local exchange service which include (1) Call Waiting ID; (2) Three-way Calling; (3) Call Forwarding; (4) Call Waiting or Talking Call WaitingSM; (5) Selective Call Forwarding, Select Call ForwardingTM or Priority Call ForwardingTM; (6) Priority Call, Priority Call RingingTM or Priority RingingTM; (7) Speed Call 8 or Speed Calling 8TM; (8) Call Screen, Call Screening, Call BlockerTM or Call Blocking; (9) Repeat DialingTM or Call AgainTM; (10) Auto RedialTM or Missed Call DialingTM; (11) Call Return or Auto Callback; and (12) PRIVACY MANAGER®.

Group B Large Package: Features associated with the provision of local exchange service which include (1) The Message Center[™], Voice Mail Plus[™], CallNotes[®] Plus, Universal CallNotes[®] or Call in One and (2) Inline[®], Pacific Bell WirePro[®], or Inside Wire Plus[™], LINE-BACKER[®], LINE-BACKER[®] w/Phone Package, LINE-BACKER[®]Basic or LINE-BACKER[®] w/Phone.

Group C Large Package: Features associated with the provision of local exchange service which include Auto Redial[™] or Missed Call Dialing[™], Call Screen[™], Call Screening, Call Blocker[™] or Call Blocking[™], Call Forwarding (includes Busy Line, Delay, Don't Answer, and Busy Line/Don't Answer), CallNotes®, The Message Center[™], Voice Mail Plus[™] or CallNotes® Plus, Voice Mail 98, Call Return[™], Auto Callback, AutoCall Block, Call Waiting, Talking Call WaitingSM, Call Waiting ID, Call Waiting ID Options, Caller ID, Inline®, Pacific Bell WirePro[™] or Inside Wire Plus[™], LINE-BACKER®, LINE-BACKER® w/Phone Package, LINE-BACKER®Basic, LINE-BACKER® w/Phone, Inline Plus, Personalized Ring/Multi Ring 1 or 2, Priority Call, Priority Call Ringing[™] or Priority Ringing[™], Remote Access to Call Forwarding, Selective Call Forwarding, Select Call Forwarding[™] or Priority Call Forwarding[™], Speed Call 8 or Speed Calling 8[™], Three-Way Calling-subscription, Internet Caller ID, PRIVACY MANAGER®, Talking Call WaitingSM, Speed Call 30, Repeat Dialing[™] or Call Again[™], VoiceMail (standard), SpeedCall 30, The Message Center (Deluxe Mailbox), Universal CallNotes® / Universal CallNotes® Plus, METRO PLAN, and Phone-ProtectSM.

Group D Package: Features associated with local exchange service which include Auto RedialTM, Busy Call Forwarding, Call Forwarding, Call Return, Call Transfer Disconnect, Call Screen, Call Waiting, Delayed Call Forwarding, Call Waiting ID, Priority Call, Priority RingingTM, PRIVACY MANAGER[®], Remote Access to Call Forwarding, Repeat Dial, Select Call ForwardingTM, 30 Member Speed Calling, Speed Calling 8TM, and Three Way Calling.

Group 1 Toll Free Access Numbers: 800-522-2020 and other Toll Free Numbers determined by the Company to be billed as a Group 1 Toll Free Access Number.

Group 2 Toll Free Access Numbers: 800-877-0000, 877-722-2141, 888-330-2323, and other Toll Free Numbers determined by the Company to be billed as a Group 2 Toll Free Access Number.

Hertz: A unit of frequency equal to one cycle per second, a standard measurement of bandwidth.

IA: Intergrated Access.

ICB: Individual Case Basis. A Service provided involving a nonstandard arrangement. The nature of such Service requirements makes it difficult or impossible to establish general Tariff provisions for such circumstances.

ILEC: Incumbent Local Exchange Carrier.

Issued: July 16, 2003

Effective: July 22, 2003

Ν

Т

Southwestern Bell Communications Services, Inc. d/b/a Southwestern Bell Long Distance d/b/a Nevada Bell Long Distance d/b/a Pacific Bell Long Distance d/b/a SBC Long Distance

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS

Inline®: A registered trademark of SBC Knowledge Ventures, Inc.

Inside Wire Plus[™]: A trademark of Southern New England Telephone Company.

InterLATA: Any call or transmission that originates in one LATA and terminates in a different LATA.

International: Involving two or more nations.

IntraLATA: Any call or transmission that originates in one LATA and terminates within the same LATA.

IOC: Interoffice Channel.

ISDN: Integrated Services Digital Network. Integrates voice, data, and video communications services via standard interfaces.

ITU: International Telecommunications Union (formerly known as the CCITT). An organization established by the United Nations with membership from virtually every government in the world. Its objectives are to set telecommunications standards and allocate frequencies to various uses.

Joint User: A corporation, association, partnership, or individual that is permitted to use a Customer's Service by mutual agreement between the Customer and the Joint User in accordance with the terms and conditions of this Tariff.

Kbps: Kilobits Per Second. One thousand Bits per second.

LATA: Local Access Transport Area. A geographically defined regulatory boundary established by the Modification of Final Judgement.

LEC: Local Exchange Carrier.

LIDB: Line Information Data Base. A data base that contains all valid telephone and calling card numbers in a region and the necessary information to perform billing validation.

LINE-BACKER®: A registered trademark of SBC Knowledge Ventures, Inc.

Local Access: The service between a subscriber's premise and a Company-designated POP.

Local Access Provider: An entity providing Local Access.

MAC: Minimum Annual Commitment.

Mbps: Mbps stands for "MegaBits Per Second" which is a million Bits per second.

METRO PLAN: A service provided by a SBC Affiliate.

Florida Tariff No. 2 11th Revised Sheet 32 Cancels 10th Revised Sheet 32

SECTION 1 - TECHNICAL TERMS AND ABBREVIATIONS

Operator Toll Assistance Services: Enable callers to place calls from their presubscribed telephone line or when away from their established primary Service location with the ability to bill the call with alternate billing options. Calls may be fully automated or may require the assistance of an operator. Operator Toll Assistance Services are also known as alternate billed services.

OTC: One Time Charge.

Oversubscription: Oversubscription allows the sum of the CIRs for all PVCs connected to an access port to exceed the speed of the access port. No individual PVC may be provisioned at a speed greater than either of its associated port speeds.

PABX: Private Automatic Branch eXchange.

Ν

Т

Т

Т

Pacific Bell Instant Office SM : A service mark of SBC Knowledge Ventures, Inc. Pacific Bell Instant Office SM is a	Ν
service provided by a SBC Affiliate.	Ν

Pacific Bell WireProTM: A trademark of SBC Knowledge Ventures, Inc.

PBX: Private Branch Exchange.

Person-to-Person: Any operator-handled call whereby the person originating a call specifies a particular person to be reached, or a particular station, room number, department, or office to be reached through a PBX attendant.

Phone-ProtectSM: Phone-ProtectSM is a service mark of SBC Knowledge Ventures, Inc.

PIC: Primary Interexchange Carrier.

PIN: Personal Identification Number. A unique number assigned to each calling card for the purpose of accessing Service.

Plexar®: A registered trademark of SBC Knowledge Ventures, Inc.

Plexar I®: A registered trademark of Southwestern Bell Telephone, L.P. Plexar I® is a service provided by a SBC N Affiliate.

POP: Point-of-Presence. A physical place at which the local telephone company terminates subscriber Circuits for long distance dial-up or leased-line communications or a Company-designated location where a facility is maintained for the purpose of providing access to the Company's Service.

Port: The physical or electrical interface through which access to the communications network is obtained.

Postalized: Charging a Flat Rate per minute irrespective of the distance the call is carried. Stems from the fact that the United States Post Office also charges a Flat Rate irrespective of how far it carries the mail (within the country).

POTS Number: Plain Old Telephone Service Number. The 10-digit telephone number associated with basic local exchange service.

Power OfficeSM : A service mark of SBC Knowledge Ventures, Inc. Power Office SM is a service provided by a SBC affiliate.	N N
PRI: Primary Rate Interface. The ISDN equivalent of a DS1 Circuit. The Primary Rate Interface consists of twenty-four 64 Kbps Channels.	
Priority Call: A service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Priority Call ForwardingTM: A trademark of Southern New England Telephone Company. Priority Call Forwarding TM is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N
Priority Call Ringing[™]: A trademark of Southern New England Telephone Company. Priority Call Ringing [™] is a service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Priority RingingTM: A trademark of Pacific Bell Telephone Company. Priority Ringing TM is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N
Priority PVCs: Priority Quality of Service offers reduced delay and packet loss between end-points when used with small, fixed-length frame traffic.	
PRIVACY MANAGER® : A registered trademark of SBC Knowledge Ventures, Inc. A service/feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	N N

Private Line: Discrete communication Bandwidth dedicated for a Customer's exclusive use. A Private Line is provisioned on facilities that may be shared and accomplished through a variety of technologies and media.

Private Line Service: Full duplex transmission/transport service between two points. Private Line Service(s) are defined by Bandwidth, signaling, media, etc.

PSTN: Public Switched Telephone Network. The worldwide voice telephone network with access to all those with telephone and access privileges.

PVC: Permanent Virtual Connection provides the customer with the electronic equivalent of a private line between two points. At the time of subscription to this form of service, a virtual circuit is established between two specific customer network addresses on the FRS network. While no physical circuits are dedicated, the two network addresses are electronically connected together.

QoS: Quality of Service.

Rate Center: A specified geographical location used for determining mileage measurements.

Remote Access to Call Forwarding: A service or feature associated with local exchange service which is defined N N in the tariff of the Affiliated LEC or Affiliated CLEC.

Repeat DialingTM: A trademark of Pacific Bell Telephone Company. Repeat DialingTM is defined in the tariff of N the Affiliated LEC.

Reseller: A Customer that resells the Company's Service(s) with the Company's authorization.

Residential Customer: A Customer whose use of the Service is primarily or substantially of a social or domestic nature; and business use, if any, is incidental.

Resp Org: Responsible Organization. The entity designated to manage and administer a Customer's SMS/800 records.

SBC: SBC Communications, Inc. The holding company of Southwestern Bell Communications Services, Inc.

SBC Phone Solution Complete for Business: A local service provided by SBC Telecom, Inc.

 SBC® Yahoo!® Dial Internet Access:
 SBC® is a registered trademark of SBC Knowledge Ventures, Inc.
 N

 Yahoo!® is a registered trademark of Yahoo! Inc.
 SBC® Yahoo!® Dial Internet Access is a service provided by a
 I

 SBC Affiliate.
 N

SEF: Severely Errored Framing event. The SE (severely errored) indicator of performance report message.

SEF Event: The occurrence of a severely errored frame. A one-second interval with eight (8) or more framing errors/framing-Bit errors.

Select Call ForwardingTM: A trademark of Pacific Bell Telephone Company. A service/feature associated with N local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC. N

Service: Any or all services provided pursuant to this Tariff.

Service Order: The standard Company order form(s), in effect from time-to-time, or Customer's forms accepted in writing by an authorized representative of the Company for Service which shall enable the Company to provide Service.

SES: Severely Errored Seconds. Errored seconds during which the error rate exceeded ten (10).

Simple Solutions®: A registered trademark of SBC Knowledge Ventures, Inc.

SimpleLinkSM: A service mark of SBC Knowledge Ventures, Inc. SimpleLinkSM is a service/feature associated N with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC. N

N

STS: Synchronous Transport Signal. A SONET electrical signal rate.

Switched Access: A transmission line that is switched through the LEC or CLEC to reach the long distance network. Switched access arrangements are only available from the subscriber's local telephone company.

Switched Services(s): Any Services that are not Data Service as defined herein which use message switches to share inter-switch transport.

T-1: Composed of twenty-four (24) 64 Kilobit Channels with a throughput capacity of 1.544 Mbps. Also called DS1.

T1C: A digital carrier facility used to transmit a DS-1 formatted digital signal at 3.152 Mbps.

Talking Call WaitingSM: A service mark of SBC Properties, Inc. Talking Call WaitingSM is a service provided by an SBC affiliate.

TFS: Toll Free Service.

The Basics®: A trademark of Southwestern Bell Telephone Company.	N
The Business Plan: A service provided by a SBC Affiliate.	N
The Message Center TM : A trademark of Pacific Bell Telephone Company.	
The Works ®: A registered trademark of Southwestern Bell Telephone Company. The Works® is service provided by a SBC affiliate.	N I
Three Way Calling: A service or feature associated with local exchange service which is defined in the tariff of the Affiliated LEC or Affiliated CLEC.	 N

Issued: July 16, 2003

Effective: July 22, 2003

Voice Mail PlusTM: A trademark of Southern New England Telephone Company.

VPN: SBC Long Distance Virtual Private Network. A Service that provides the functionality and capabilities of a private network through the use of shared transmission facilities.

VPN Authorization Code: A code used to override CSR restrictions imposed on the VPN Service.

VPN On-Net: A feature that enables the Customer or user to dial from one VPN member station to another within the Customer's VPN service.

VPN Off-Net: A feature that enables the Customer or user to dial from a VPN member station to a number outside of the Customer's VPN network. VPN Off-Net calling may be initiated from a switched, dedicated, or remote access location.

VRA: The authentication code and PIN used when making a VPN remote access call.

VRA Number: VPN Remote Access Toll Free Number.

WATS: Wide Area Telecommunications Service.

Ν

Wire Center: A specified geographical location used for determining mileage measurements.

WTN: Working Telephone Number.

Ν

Ν

SECTION 2 - RULES AND REGULATIONS

2.26 Revenue and Term Plan Commitments

2.26.1 General

- (A) As a condition of obtaining a specific Service offering or a specific optional pricing plan, a Customer may be required to make a (1) MAC and a term plan commitment; (2) an MMC and a term plan commitment; or (3) an MMC without a term plan commitment. The terms and conditions for qualifying for each specific offering is described in Section 3 this Tariff. Business Customers subscribing to one of the Company's High Volume Calling plans or SBC Long Distance Virtual Private Network (VPN) are required to sign term plan agreements. Business Customers subscribing to any other Service offering may make a verbal MMC, MAC or term plan commitments.
- (B) By committing to a MAC or an MMC, the Customer commits to spending a predetermined dollar revenue volume, either annually in the case of a MAC or monthly in the case of an MMC.
- (C) By making a term plan commitment, the Customer commits to remain a Customer of Company for a specified length of time.

Issued: July 16, 2003

Effective: July 22, 2003

2.26 Revenue and Term Plan Commitments (continued)

2.26.2 Calculation of MAC and MMC

 (B) Customer Subscribes To Business Long Distance 50, Business Long Distance 75, Business Long Distance 100, Business Long Distance 200, Business Long Distance Value 50, Business Long Distance Value 100, Business Long Distance 50 Connections 1 Service, Business Long Distance
 50 Connections 2 Service, Business Long Distance 50 Connections 3
 Service, Business Long Distance 100 Connections 1 Service, Business
 Long Distance 100 Connections 2 Service, Long Distance 100 Connections
 3 Service, Business Long Distance 50 Connections 1 Plus, Business Long
 Distance 50 Connections 2 Plus, Business Long Distance 100 Connections
 N

For Customers subscribing to any of the Services listed above, only the revenue associated with the Service and Calling Card - Option 2 contributes T towards meeting the MMC. If a Customer subscribes to other switched toll T free services, outbound or calling card Services, the revenue will not be counted when calculating whether or not the Customer has met the MMC T commitment associated with the Service.

- .1 An MMC commits the Customer to paying the Company a predetermined amount of revenue resulting from:
 - .a 1+ outbound domestic and International usage;
 - .b domestic switched TFS usage and usage charges associated with Canadian Toll Free Service;

Issued: July 16, 2003

Effective: July 22, 2003

Т

2.26 Revenue and Term Plan Commitments (continued)

- 2.26.2 Calculation of MAC and MMC (continued)
 - Customer Subscribes To Business Long Distance 50, Business Long **(B)** Т Distance 75, Business Long Distance 100, Business Long Distance 200, Ν Business Long Distance Value 50, Business Long Distance Value 100, Business Long Distance 50 Connections 1 Service, Business Long Distance 50 Connections 2 Service, Business Long Distance 50 Connections 3 Service, Business Long Distance 100 Connections 1 Service, Business Long Distance 100 Connections 2 Service, Long Distance 100 Connections 3 Service, Business Long Distance 50 Connections 1 Plus, Business Long Distance 50 Connections 2 Plus, Business Long Distance 100 Connections 1 1 Plus, or Business Long Distance 100 Connections 2 Plus (continued) N
 - .1 (continued)
 - .c domestic and International usage for calling card calls billed to the Calling Card Option 2; and
 - .d any credits associated with a qualified usage item.

Issued: July 16, 2003

Effective: July 22, 2003

2.26 Revenue and Term Plan Commitments (continued)

2.26.2 Calculation of MAC and MMC (continued)

- (B) Customer Subscribes To Business Long Distance 50, Business Long Distance 75, Business Long Distance 100, Business Long Distance 200, Business Long Distance Value 50, Business Long Distance Value 100, Business Long Distance 50 Connections 1 Service, Business Long Distance 50 Connections 3 Service, Business Long Distance 100 Connections 1 Service, Business Long Distance 100 Connections 3 Service, Business Long Distance 50 Connections 2 Service, Long Distance 100 Connections 3 Service, Business Long Distance 50 Connections 1 Plus, Business Long Distance 50 Connections 1 Plus, or Business Long Distance 100 Connections 2 Plus (continued)
 - .2 Charges associated with directory assistance Service, monthly recurring charges and one time charges, taxes and surcharges, reductions because of promotions (free minutes or reduced price per minute), and adjustments that are not associated with a particular usage item are not included in determining whether the Customer has met the MMC or MAC.
- (C) Customer Subscribes To Business Domestic Saver 15 Plus 1 Year, Business Domestic Saver 15 Plus 2 Year, Business Domestic Saver 15, Business Domestic Saver 15 Deluxe, Business Domestic Saver, Business Domestic Saver Deluxe, Business Domestic Value Saver 15, Business Domestic Saver 15 Connections 1 Service, Business Domestic Saver 15 Connections 2 Service or Business Domestic Saver 15 Connections 3 Service, Business Domestic Saver 15 Connections 1 Plus, or Business Domestic Saver 15 Connections 2 Plus

For Customers subscribing to any of the Services listed above, only revenue associated with the Service and Calling Card - Option 2 contributes towards meeting the MMC. If a Customer subscribes to other switched toll free services, outbound or calling card Services, the revenue will not be counted when calculating whether or not the Customer has met the MMC associated with the Service.

Т

N

N

Т

Ť

Т

Т

Т

Т

Issued: July 16, 2003

Effective: July 22, 2003

- 2.26 Revenue and Term Plan Commitments (continued)
 - 2.26.2 Calculation of MAC and MMC (continued)
 - (C) Customer Subscribes To Business Domestic Saver 15 Plus 1 Year, Business T
 Domestic Saver 15 Plus 2 Year, Business Domestic Saver 15, Business |
 Domestic Saver 15 Deluxe, Business Domestic Saver, Business Domestic
 Saver Deluxe, Business Domestic Value Saver 15, Business Domestic
 Saver 15 Connections 1 Service, Business Domestic Saver 15 Connections
 Service or Business Domestic Saver 15 Connections 3 Service, Business
 Domestic Saver 15 Connections 1 Plus, or Business Domestic Saver 15
 Connections 2 Plus (continued)
 - .1 An MMC commits the Customer to paying the Company a predetermined amount of revenue resulting from:
 - .a 1+ outbound domestic usage;
 - .b domestic switched TFS usage and usage charges associated T with Canadian Toll Free Service;
 - .c domestic usage for fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card -Option 2; and
 - .d any credits associated with a qualified usage item.

Issued: July 16, 2003

Effective: July 22, 2003

2.26 Revenue and Term Plan Commitments (continued)

- 2.26.2 Calculation of MAC and MMC (continued)
 - (C) Customer Subscribes To Business Domestic Saver 15 Plus 1 Year, Business T
 Domestic Saver 15 Plus 2 Year, Business Domestic Saver 15, Business |
 Domestic Saver 15 Deluxe, Business Domestic Saver, Business Domestic |
 Saver Deluxe, Business Domestic Value Saver 15, Business Domestic |
 Saver 15 Connections 1 Service, Business Domestic Saver 15 Connections |
 Service or Business Domestic Saver 15 Connections 3 Service, Business |
 Domestic Saver 15 Connections 1 Plus, or Business Domestic Saver 15 |
 Connections 2 Plus (continued)
 - .2 Charges associated with directory assistance Service, monthly recurring charges and one time charges, taxes and surcharges, reductions because of promotions (free minutes or reduced price per minute), and good will adjustments that are not associated with a particular usage item are not included in determining whether the Customer has met the MMC.

Issued: July 16, 2003

Effective: July 22, 2003

- 2.26 Revenue and Term Plan Commitments (continued)
 - 2.26.3 Under-Utilization Charges
 - (A) Unmet MAC

If a Customer subscribing to any of the Company's High Volume Calling plans, the Customer will be billed the difference between the actual usage and the unmet MAC within two (2) billing cycles of the Customer's yearly anniversary date. If a Customer subscribing to VPN Service fails to meet its MAC, the Customer will be billed the difference between the actual usage and the unmet MAC as an under-utilization charge.

- (B) Unmet MMC
 - .1 If a Customer subscribing to any of the Company's High Volume Calling plans fails to meet its MMC in any given billing month, the Customer will be billed the difference between the actual usage revenue and the unmet MMC within two (2) billing cycles of the billing period in which the shortfall occurred. If a Customer subscribing to any of the following Services fails to meet its MMC in any given billing month, the Customer will be billed the difference between the actual usage revenue and the unmet MMC for the billing cycle in which the shortfall occurred.

.a	Business Domestic Saver	N/M
.b	Business Domestic Saver 15	
.c	Business Domestic Saver 15 Connections 1 Service	ĺ
.d	Business Domestic Saver 15 Connections 2 Service	ĺ
.e	Business Domestic Saver 15 Connections 3 Service	
.f	Business Domestic Saver 15 Deluxe	j
.g	Business Domestic Saver Deluxe	į
.g .h	Business Long Distance 50	ĺ
.i	Business Long Distance 50 Connections 1 Service	
	Business Long Distance 50 Connections 2 Service	i
.j .k	Business Long Distance 50 Connections 3 Service	İ
.1	Business Long Distance 100	N/M
	5	

M - Material moved to Original Sheet 98.1

Issued: July 16, 2003

Effective: July 22, 2003

Т

SECTION 2 - RULES AND REGULATIONS Ν 2.26 Revenue and Term Plan Commitments (continued) 2.26.3 Under-Utilization Charges (continued) Unmet MMC (continued) **(B) Business Long Distance 100 Connections 1 Service** .m Business Long Distance 100 Connections 2 Service .n Business Long Distance 100 Connections 3 Service .0 **Business Long Distance 200** .p Business Domestic Saver 15 Connections 1 Plus .q **Business Domestic Saver 15 Connections 2 Plus** .r **Business Long Distance 50 Connections 1 Plus** .s Business Long Distance 50 Connections 2 Plus .t Business Long Distance 100 Connections 1 Plus .u Business Long Distance 100 Connections 2 Plus .v **Business Domestic Saver 15 Plus** .W Reserved for future use .x Reserved for future use .y **Business Long Distance Value 50** .Z Business Long Distance Value 100 .aa **Business Domestic Value Saver 15** .ab Business Domestic Saver 15 Connections 1 Plus Service .ac Business Long Distance 50 Connections 1 Plus Service .ad Business Long Distance 100 Connections 1 Plus Service .ae Business Domestic Saver 15 Connections 2 Plus Service .af Business Long Distance 50 Connections 2 Plus Service .ag Business Long Distance 100 Connections 2 Plus Service Ň .ah Customers subscribing to any of the Company's High Volume Calling М .2 plans and committing to an MMC will be given up to a three (3) month period for usage ramp up before any under-utilization charge is assessed. If a Customer subscribing to any of the Company's High Volume Calling plans subscribes to an MMC on any date other than the first day of the billing cycle, the partial first month is counted as a full month when determining the length of the ramp up period. M

M - Material moved from 11th Revised Sheet 98

Issued: July 16, 2003

Effective: July 22, 2003

Т

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.18 Business Domestic Saver 15 Connections 3 Service

- (A) Business Domestic Saver 15 Connections 3 Service is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
 - .3 subscribe to and maintain the following products or services provided by an SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo®! dial-up Internet access or TI Integrated Access or dedicated web hosting; and (2) Cingular wireless service; and (3) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic
 Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or
 Ameritech Centrex Service (ACS) service package as defined in SBC Affiliate:

3.7	Custon	n Business Services (continued)				
	3.7.18	Busi	siness Domestic Saver 15 Connections 3 Service (continued)			
		(A)	(con	ntinued)		
			.3	(conti	nued)	
				.a	a minimum of one business access line, and,	
				.b	an inside wire maintenance product associated with each business access line, and,	
				.c	at least one instance of Caller ID, and,	
				.d	at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.	
			.4	subsc	onstrate to the satisfaction of the Company at the time of cribing to the Service that the Customer also subscribes to the act or service described in Section 3.7.18 (A).3 of this Tariff.	

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.18 Business Domestic Saver 15 Connections 3 Service (continued)
 - (A) (continued)
 - .5 except as described below, associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.18 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 3 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 3 Service will be billed.
 - .6 commit to
 - an MMC of \$15 per month for a 1-year term plan or
 - an MMC of \$15 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

N

N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.18 Business Domestic Saver 15 Connections 3 Service (continued)
 - (B) The Customer may subscribe to Business Domestic Saver 15 Connections 3 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.18 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on whether the Customer subscribes to a 1-year or 2-year term plan.

- 3.7 Custom Business Services (continued)
 - 3.7.18 Business Domestic Saver 15 Connections 3 Service (continued)
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.18 of this Tariff.
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.18 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 3 Service and will be moved to Business Domestic Saver 15 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.18 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connection 3, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

N

Т

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.19 Business Long Distance 50 Connections 3 Service

- (A) Business Long Distance 50 Connections 3 Service is a custom combination N switched TFS, outbound, and calling card Flat Rate Service available to |
 Business Customers that: |
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; and N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.19 Business Long Distance 50 Connections 3 Service
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by an SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo®! dial-up Internet access or TI Integrated Access or dedicated web hosting; and (2) Cingular wireless service; and (3) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLinkSM, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.19 Business Long Distance 50 Connections 3 Service (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the product or service described in Section 3.7.19 (A).3 of this Tariff.
 - .5 except as described below, associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.19 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 3 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 3 Service will be billed.
 - .6 commit to
 - an MMC of \$50 per month for a 1-year term plan or
 - an MMC of \$50 per month for a 2-year term plan and sign a written term plan agreement with the Company.

Ň

N

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (cont

- 3.7.19 Business Long Distance 50 Connections 3 Service (continued)
 - (A) (continued)

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

- (B) The Customer may subscribe to Business Long Distance 50 Connections 3 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.19 (A).3 of this Tariff.
- (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7.19 Business Long Distance 50 Connections 3 Service (continued)
 - (D) The Customer's usage rate for each call is based on whether the Customer subscribes to Service on a 1-year or 2-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.19 of this Tariff.
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.19 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 3 Service and will be moved to Business Long Distance 50 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.19 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 3 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

3.7 Custom Business Services (continued)

3.7.20	Business Long Distance 100 Connections 3 Service	Т

- (A) Business Long Distance 100 Connections 3 Service is a custom
 N combination switched TFS, outbound, and calling card Flat Rate Service
 available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; and N

- 3.7 Custom Business Services (continued)
 - 3.7.20 Business Long Distance 100 Connections 3 Service
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by an SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo®! dial-up Internet access or TI Integrated Access or dedicated web hosting; and (2) Cingular wireless service; and (3) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

Ň

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.20 Business Long Distance 100 Connections 3 Service
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.20 (A).3 of this Tariff.
 - .5 except as described below, associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.20 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 3 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 3 Service will be billed.
 - .6 commit to
 - an MMC of \$100 per month for a 1-year term plan or
 - an MMC of \$100 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff.

3.7	Custon	m Business Services (continued)				
	3.7.20	Business Long Distance 100 Connections 3 Service (continued)				
		(A)	(continued)	1		
			This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.			
		(B)	The Customer may subscribe to Business Long Distance 100 Connections 3 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.20 (A).3 of this Tariff.			
		(C)	Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.			
		(D)	For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.	 N		

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.20 Business Long Distance 100 Connections 3 Service (continued)

- (E) The Customer's usage rate for each call is based on whether the Customer subscribes to Service on a 1-year or 2-year term plan.
- (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.13 of this Tariff.
- (G) If the Customer fails to maintain the required products or services described in Section 3.7.20 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 3 Service and will be moved to Business Long Distance 100 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.20 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 3 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

Т

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.21 Business Domestic Saver 15 Connections 2 Service

- (A) Business Domestic Saver 15 Connections 2 Service is a custom
 N combination switched TFS, outbound, and calling card Flat Rate Service
 available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; N

- 3.7 Custom Business Services (continued)
 - 3.7.21 Business Domestic Saver 15 Connections 2 Service
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by an SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo®! dial-up Internet access or TI Integrated Access or dedicated web hosting and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

SECTION 3 - DESCRIPTION OF SERVICES

	3.7	Custom	Business	Services	(continued)
--	-----	--------	-----------------	----------	-------------

- 3.7.21 Business Domestic Saver 15 Connections 2 Service (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.21 (A).3 of this Tariff.
 - .5 except as described below, associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.21 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 2 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 2 Service will be billed.

Issued: July 16, 2003

3.7

SECTION 3 - DESCRIPTION OF SERVICES

Custom Business Services (continued)				
3.7.21 Business Domestic Saver 15 Connections 2 Service (continued)				
(A	(continued)			
	 .6 commit to an MMC of \$15 per month for a 1-year term plan or an MMC of \$15 per month for a 2-year term plan and sign a written term plan agreement with the Company. For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff. 			
	This Service is established at the BTN level and is only available for asingle BTN. Service is available for Customers that commit to a 1-year or2-year term plan.			

- 3.7 Custom Business Services (continued)
 - 3.7.21 Business Domestic Saver 15 Connections 2 Service (continued)
 - (B) The Customer may subscribe to Business Domestic Saver 15 Connections 2 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.21 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on whether the Customer subscribes to a 1-year or 2-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.21 of this Tariff.

N

N

- 3.7 Custom Business Services (continued)
 - 3.7.21 Business Domestic Saver 15 Connections 2 Service (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.21 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 2 Service and will be moved to Business Domestic Saver 15 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.21 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 2 Service , the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

Т

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.22 Business Long Distance 50 Connections 2 Service

- (A) Business Long Distance 50 Connections 2 Service is a custom combination N switched TFS, outbound, and calling card Flat Rate Service available to |
 Business Customers that: |
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.22 Business Long Distance 50 Connections 2 Service
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by an SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo®! dial-up Internet access or TI Integrated Access or dedicated web hosting and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.22 Business Long Distance 50 Connections 2 Service (continued)

- (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.22 (A).3 of this Tariff.
 - .5 except as described below, associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.22 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 2 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 2 Service will be billed.
 - .6 commit to
 - an MMC of \$50 per month for a 1-year term plan or
 - an MMC of \$50 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

- 3.7 Custom Business Services (continued)
 - 3.7.22 Business Long Distance 50 Connections 2 Service (continued)
 - (B) The Customer may subscribe to Business Long Distance 50 Connections 2 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.22 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on whether the Customer subscribes to Service on a 1-year or 2-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.22 of this Tariff.

Ν

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.22 Business Long Distance 50 Connections 2 Service (continued)

(F) If the Customer fails to maintain the required products or services described in Section 3.7.22 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 2 Service and will be moved to Business Long Distance 50 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.22 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 2 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

3.7 Custom Business Services (continued)

3.7.23 Business Long Distance 100 Connections 2 Service T

- (A) Business Long Distance 100 Connections 2 Service is a custom
 N combination switched TFS, outbound, and calling card Flat Rate Service
 available to Business Customers that:
 - .1 request to be provisioned under this optional Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; and N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.23 Business Long Distance 100 Connections 2 Service
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by an SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo®! dial-up Internet access or TI Integrated Access or dedicated web hosting and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.23 Business Long Distance 100 Connections 2 Service
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.23 (A).3 of this Tariff.
 - .5 except as described below, associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.23 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 2 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 2 Service will be billed.
 - .6 commit to
 - an MMC of \$100 per month for a 1-year term plan or
 - an MMC of \$100 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

N

Ν

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.23 Business Long Distance 100 Connections 2 Service (continued)
 - (B) The Customer may subscribe to Business Long Distance 100 Connections 2 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.23 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.
 - (E) The Customer's usage rate for each call is based on whether the Customer subscribes to Service on a 1-year or 2-year term plan.

- 3.7 Custom Business Services (continued)
 - 3.7.23 Business Long Distance 100 Connections 2 Service (continued)
 - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.13 of this Tariff.
 - (G) If the Customer fails to maintain the required products or services described in Section 3.7.23 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 2 Service and will be moved to Business Long Distance 100 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.23 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 2 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

3.7 Custom Business Services (continued)

3.7.24 Business Domestic Saver 15 Connections 1 Service T

- (A) Business Domestic Saver 15 Connections 1 Service is a custom N
 combination switched TFS, outbound, and calling card Flat Rate Service
 available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS;

Ν

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.24 Business Domestic Saver 15 Connections 1 Service (continued)
 - (A) (continued)
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

N

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)	3.	7	Custom	Business	Services	(continued)
------------------------------------------	----	---	--------	-----------------	----------	-------------

3.7.24 Business Domestic Saver 15 Connections 1 Service (continued)

- (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.24 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.24 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 1 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 1 Service will be billed; and

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7	Custom	Business	Services	(continued)

- 3.7.24 Business Domestic Saver 15 Connections 1 Service (continued)
 - (A) (continued)
 - .6 commit to
 - an MMC of \$15 per month for a 1-year term plan or
 - an MMC of \$15 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

(B) The Customer may subscribe to Business Domestic Saver 15 Connections 1 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.24 (A).3 of this Tariff.

Issued: July 16, 2003

3.7

		SECTION 3 - DESCRIPTION OF SERVICES			
Custon	n Busi	ness Services (continued)	N		
3.7.24	Business Domestic Saver 15 Connections 1 Service (continued)				
	(C)	Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.			
	(D)	The Customer's usage rate for each call is based on whether the Customer subscribes to a 1-year or 2-year term plan.			
	(E)	Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.24 of this Tariff.	 N		

3.7 Custom Business Services (continued)

3.7.24 Business Domestic Saver 15 Connections 1 Service (continued)

(F) If the Customer fails to maintain the required products or services described in Section 3.7.24 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 1 Service and will be moved to Business Domestic Saver 15 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.24 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 1 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

Ν

Т

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.25 Business Long Distance 50 Connections 1 Service

- (A) Business Long Distance 50 Connections 1 Service is a custom combination N switched TFS, outbound, and calling card Flat Rate Service available to
 Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; N

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.25 Business Long Distance 50 Connections 1 Service
 - (A) (continued)
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.25 Business Long Distance 50 Connections 1 Service (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.25 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.25 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 1 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 1 Service will be billed; and

N

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

- 3.7.25 Business Long Distance 50 Connections 1 Service (continued)
 - (A) (continued)
 - .6 commit to
 - an MMC of \$50 per month for a 1-year term plan or
 - an MMC of \$50 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

- (B) The Customer may subscribe to Business Long Distance 50 Connections 1 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.25 (A).3 of this Tariff.
- (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.

N

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.25 Business Long Distance 50 Connections 1 Service (continued)

- (D) The Customer's usage rate for each call is based on whether the Customer subscribes to Service on a 1-year or 2-year term plan.
- (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.25 of this Tariff.
- (F) If the Customer fails to maintain the required products or services described in Section 3.7.25 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 1 Service and will be moved to Business Long Distance 50 for the same term as the Customer's current term plan associated with this Service unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.25 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 1 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

Т

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.26 Business Long Distance 100 Connections 1 Service

- (A) Business Long Distance 100 Connections 1 Service is a custom N
 combination switched TFS, outbound, and calling card Flat Rate Service |
 available to Business Customers that: |
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS; N

N

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.26 Business Long Distance 100 Connections 1 Service
 - (A) (continued)
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from an SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as described in Section 1 of this Tariff from a SBC Affiliate.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.26 Business Long Distance 100 Connections 1 Service
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.26 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of an Affiliated LEC or Affiliated CLEC that are described in Section 3.7.26 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 1 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 1 Service will be billed; and

- 3.7 Custom Business Services (continued)
 - 3.7.26 Business Long Distance 100 Connections 1 Service (continued)
 - (A) (continued)
 - .6 commit to
 - an MMC of \$100 per month for a 1-year term plan or
 - an MMC of \$100 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is available for Customers that commit to a 1-year or 2-year term plan.

- (B) The Customer may subscribe to Business Long Distance 100 Connections 1 Service for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.26 (A).3 of this Tariff.
- (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.26 Business Long Distance 100 Connections 1 Service (continued)
 - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.
 - (E) The Customer's usage rate for each call is based on whether the Customer subscribes to Service on a 1-year or 2-year term plan.
 - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.13 of this Tariff.
 - (G) If the Customer fails to maintain the required products or services described in Section 3.7.26 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 1 Service and will be moved to Business Long Distance 100 unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.26 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 1 Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

3.7	Custom Business Services (continued)	Ν
	3.7.33 Reserved for future use.	Ν

3.7	Custom Business Services (continued)	Ν
	3.7.34 Reserved for future use.	Ν

N

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

- 3.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year
 - (A) Business Domestic Saver 15 Connections 1 Plus Service 1 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.35 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.35 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 1 Plus Service 1 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 1 Plus Service 1 Year will be billed; and
 - .6 commit to an MMC of \$15 per month for a 1-year term plan.

For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

- 3.7 Custom Business Services (continued)
 - 3.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year (continued)
 - (B) The Customer may subscribe to Business Domestic Saver 15 Connections 1 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.35 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on the 1-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.35 of this Tariff.

Ν

Ν

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.35 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 1 Plus Service 1 Year and will be moved to Business Domestic Saver 15 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.35 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 1 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Domestic Saver 15 Connections 1 Service for an additional 1-year term, as described in Section 3.7.24 of this tariff, unless otherwise specified by the Customer. 3.7

SECTION 3 - DESCRIPTION OF SERVICES Ν Custom Business Services (continued) 3.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year Business Long Distance 50 Connections 1 Plus Service 1 Year is a custom (A) combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that: .1 request to be provisioned under this Service; .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS: subscribe to a minimum of two Centrex station lines from a SBC .3 Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate: a minimum of one business access line, and, .a

- .b an inside wire maintenance product associated with each business access line, and,
- .c at least one instance of Caller ID, and,
- .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

- 3.7 Custom Business Services (continued)
 - 3.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.36 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.36 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 1 Plus Service 1 Year. Exceptions to this requirement are:
 (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 1 Plus Service 1 Year will be billed; and
 - .6 commit to an MMC of \$50 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 50 Connections 1 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.36 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on a 1-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.36 of this Tariff.

Ν

Ň

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

- 3.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.36 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 1 Plus Service 1 Year and will be moved to Business Long Distance 50 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7 9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.36 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 1 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 50 Connections 1 Service for an additional 1-year term, as described in Section 3.7.25 of this tariff, unless otherwise specified by the Customer.

Issued: July 16, 2003

3.7

Ν

N

Custon	1 Busii	ness S	ervices (continued)
3.7.37	Busir	ness L	ong Distance 100 Connections 1 Plus Service 1 Year
	(A)	com	ness Long Distance 100 Connections 1 Plus Service 1 Year is a custom bination switched TFS, outbound, and calling card Flat Rate Service lable to Business Customers that:
		.1	request to be provisioned under this Service;
		.2	utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;

- 3.7 Custom Business Services (continued)
 - 3.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year
 - (A) (continued)
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - d at least one instance of each of any three call control features from Group D Package as described in Section 1 of this Tariff from a SBC Affiliate.

3.7

SECTION 3 - DESCRIPTION OF SERVICES Ν Custom Business Services (continued) 3.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year (A) (continued) .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.37 (A).3 of this Tariff; except as described below associate the billing for the products or .5 services of a SBC Affiliate that are described in Section 3.7.37 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 1 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 1 Service will be billed; and .6 commit to an MMC of \$100 per month for a 1-year term.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

- 3.7 Custom Business Services (continued)
 - 3.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 100 Connections 1 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.37 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.
 - (E) The Customer's usage rate for each call is based on a 1-year term plan.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year (continued)
 - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.37 of this Tariff.
 - (G) If the Customer fails to maintain the required products or services described in Section 3.7.37 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 1 Plus Service 1 Year and will be moved to Business Long Distance 100 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.37 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 1, Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 100 Connections 1 Service for an additional 1-year term, as described in Section 3.7.26 of this tariff, unless otherwise specified by the Customer.

			c L	SECTION 3 - DESCRIPTION OF SERVICES	
3.7	Custon	n Busi	ness Se	ervices (continued)	N
	3.7.38	Busin	ness Do	omestic Saver 15 Connections 1 Plus Service 2 Year	
		(A)	comb	ness Domestic Saver 15 Connections 1 Plus Service 2 Year is a custom bination switched TFS, outbound, and calling card Flat Rate Service able to Business Customers that:	
			.1	request to be provisioned under this Service;	
			.2	utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;	 N

- 3.7 Custom Business Services (continued)
 - 3.7.38 Business Domestic Saver 15 Connections 1 Plus Service 2 Year (continued)
 - (A) (continued)
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.38 Business Domestic Saver 15 Connections 1 Plus Service 2 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.38 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.38 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 1 Plus Service 2 Year. Exceptions to this requirement are:
 (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 1 Plus Service 2 Year will be billed; and
 - .6 commit to an MMC of \$15 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 2-year term plan.

3.7 Custom Business Services (continue

- 3.7.38 Business Domestic Saver 15 Connections 1 Plus Service 2 Year (continued)
 - (B) The Customer may subscribe to Business Domestic Saver 15 Connections 1 Plus Service 2 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.38 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on the 2-year term plan.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.38 Business Domestic Saver 15 Connections 1 Plus Service 2 Year (continued)
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.38 of this Tariff.
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.38 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 1 Plus Service 2 Year and will be moved to Business Domestic Saver 15 for an additional 2-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.38 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 1 Plus Service 2 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 2-year term, the Customer will be moved to Business Domestic Saver 15 Connections 1 Service for an additional 2-year term, as described in Section 3.7.24 of this tariff, unless otherwise specified by the Customer.

3.7	Custom Business Services (continued)

- 3.7.39 Business Long Distance 50 Connections 1 Plus Service 2 Year
 - (A) Business Long Distance 50 Connections 1 Plus Service 2 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS; N

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.39 Business Long Distance 50 Connections 1 Plus Service 2 Year
 - (A) (continued)
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.39 Business Long Distance 50 Connections 1 Plus Service 2 Year(continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.39 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.39 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 1 Plus Service 2 Year. Exceptions to this requirement are:
 (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Long Distance 50 Connections 1 Plus Service 2 Year will be billed; and
 - .6 commit to an MMC of \$50 per month for a 2-year term plan and sign a written agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 2-year term plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.39 Business Long Distance 50 Connections 1 Plus Service 2 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 50 Connections 1 Plus Service 2 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.39 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on a 2-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.39 of this Tariff.

3.7 Custom Business Services (continued)

- 3.7.39 Business Long Distance 50 Connections 1 Plus Service 2 Year (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.39 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 1 Plus Service 2 Year and will be moved to Business Long Distance 50 for an additional 2-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.39 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 1 Plus 2 Year Service 2 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 2-year term, the Customer will be moved to Business Long Distance 50 Connections 1 Service for an additional 2-year term, as described in Section 3.7.25 of this tariff, unless otherwise specified by the Customer.

N

3.7	Custom Business Services (continued)

- 3.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year
 - (A) Business Long Distance 100 Connections 1 Plus Service 2 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; N

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year
 - (A) (continued).
 - .3 subscribe to a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff, or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as described in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

- 3.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.40 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.40 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 1 Service. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 1 Service will be billed; and
 - .6 commit to an MMC of \$100 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 2-year term plan.

Ň

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 100 Connections 1 Plus Service 2 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.40 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year (continued)
 - (E) The Customer's usage rate for each call is based on a 2-year term plan.
 - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.40 of this Tariff.
 - (G) If the Customer fails to maintain the required products or services described in Section 3.7.40 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 1 Plus Service 2 Year and will be moved to Business Long Distance 100 for an additional 2-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.40 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 1 Service 2 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 2-year term, the Customer will be moved to Business Long Distance 100 Connections 1 Service for an additional 2-year term, as described in Section 3.7.26 of this tariff, unless otherwise specified by the Customer.

3.7	Custor	n Business Services (continued)			
	3.7.41	Business Domestic Saver 15 Connections 2 Plus Service, 1 Year			
		(A)	com	Business Domestic Saver 15 Connections 2 Plus Service 1 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:	
			.1	request to be provisioned under this Service;	
			.2	utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;	 N

N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate;

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.41 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.41 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 2 Plus Service 1 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 2 Plus Service 1 Year will be billed; and
 - .6 commit to an MMC of \$15 per month for a 1-year term.

For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)
 - (B) The Customer may subscribe to Business Domestic Saver 15 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.41 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based a 1-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.41 of this Tariff.

- 3.7 Custom Business Services (continued)
 - 3.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.41 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 2 Plus Service 1 Year and will be moved to Business Domestic Saver 15 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.41 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 2 Plus Service, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Domestic Saver 15 Connections 2 Service for an additional 1-year term, as described in Section 3.7.21 of this tariff, unless otherwise specified by the Customer.

Ν

3.7 Custom Business Services (continued)	
------------------------------------------	--

- 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year
 - (A) Business Long Distance 50 Connections 2 Plus Service 1 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance | network for outbound calling and/or utilize Switched | Access to receive calls from the long distance network N for TFS;

- 3.7 Custom Business Services (continued)
 - 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.42 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.42 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 2 Plus Service 1 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 2 Plus Service 1 Year will be billed; and
 - .6 commit to an MMC of \$50 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

N

- 3.7 Custom Business Services (continued)
 - 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 50 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.42 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based a 1-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.42 of this Tariff.

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

- 3.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.42 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 50 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.42 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 50 Connections 2 Service for an additional 1-year term, as described in Section 3.7.22 of this tariff, unless otherwise specified by the Customer.

		S	SECTION 3 - DESCRIPTION OF SERVICES		
3.7	Custom Business Services (continued)				
	3.7.43 Bus	iness L	ong Distance 100 Connections 2 Plus Service 1 Year		
	(A)	com	Business Long Distance 100 Connections 2 Plus Service 1 Year is a custom ombination switched TFS, outbound, and calling card Flat Rate Service vailable to Business Customers that:		
		.1	request to be provisioned under this optional Service;		
		.2	utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS; and	 N	

- 3.7 Custom Business Services (continued)
 - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.43 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.43 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 2 Plus Service. Exceptions to this requirement are:
 (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 2 Plus Service 1 Year will be billed; and
 - .6 commit to an MMC of \$100 per month for a 1-year term

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 1-year term plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 100 Connections 2 Plus Service 1 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.43 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year (continued)
 - (E) The Customer's usage rate for each call is based a 1-year term plan.
 - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.43 of this Tariff.
 - (G) If the Customer fails to maintain the required products or services described in Section 3.7.43 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 2 Plus Service 1 Year and will be moved to Business Long Distance 100 for an additional 1-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.43 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 2 Plus Service 1 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 1-year term, the Customer will be moved to Business Long Distance 100 Connections 2 Service for an additional 1-year term, as described in Section 3.7.23, unless otherwise specified by Customer.

3.7	Custom	Business	Services ((continued)

- 3.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year
 - (A) Business Domestic Saver 15 Connections 2 Plus Service 2 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS;

- 3.7 Custom Business Services (continued)
 - 3.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.;

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.44 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.44 (A).3 of this Tariff with the Customer's BTN for Business Domestic Saver 15 Connections 2 Plus Service 2 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Domestic Saver 15 Connections 2 Plus Service 2 Year will be billed; and

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year(continued)

- (A) (continued)
 - .6 commit to an MMC of \$15 per month for a 2-year term plan and sign a written term agreement with the Company.

For rules and regulations regarding MMCs and term plans, see Section 2.26 of this Tariff.

This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 2-year term plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year (continued)
 - (B) The Customer may subscribe to Business Domestic Saver 15 Connections 2 Plus Service 2 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.44 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based on a 2-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.44 of this Tariff.

- 3.7 Custom Business Services (continued)
 - 3.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year (continued)
 - (F) If the Customer fails to maintain the required products or services described in Section 3.7.44 (A).3 of this Tariff, the Customer will no longer qualify for Business Domestic Saver 15 Connections 2 Plus Service 2 Year and will be moved to Business Domestic Saver 15 for an additional 2-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Domestic Saver 15, the rates and charges in Section 4.7.11 of the Tariff will apply in lieu of the rates and charges in Section 4.7.44 of this Tariff.

If the Customer is moved to Business Domestic Saver 15 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Domestic Saver 15 Connections 2 Plus Service 2 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 2-year term, the Customer will be moved to Business Domestic Saver 15 Connections 2 Service for an additional 2-year term, as specified in Section 3.7.21 of this tariff, unless otherwise specified by the Customer.

Ν

N

3.7	Custor	n Busi	ness S	ss Services (continued)		
	3.7.45	Busi	ong Distance 50 Connections 2 Plus Service 2 Year			
		(A)	com	Business Long Distance 50 Connections 2 Plus Service 2 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:		
			.1	request to be provisioned under this Service;		
			.2	utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;	 	

N

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.45 Business Long Distance 50 Connections 2 Plus Service 2 Year
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.45 Business Long Distance 50 Connections 2 Plus Service 2 Year (continued)
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.45 (A).3 of this Tariff;
 - .5 except as described below associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.45 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 50 Connections 2 Plus Service 2 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 50 Connections 2 Plus Service 2 Year will be billed; and
 - .6 commit to an MMC of \$50 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 2-year term plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.45 Business Long Distance 50 Connections 2 Plus Service 2 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 50 Connections 2 Plus Service 2 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.45 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) The Customer's usage rate for each call is based a 2-year term plan.
 - (E) Outbound calls, TFS calls, and fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.45 of this Tariff.

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.45 Business Long Distance 50 Connections 2 Plus Service 2 Year (continued)

(F) If the Customer fails to maintain the required products or services described in Section 3.7.45 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 50 Connections 2 Plus Service 2 Year and will be moved to Business Long Distance 50 for an additional 2-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 50, the rates and charges in Section 4.7.9 of the Tariff will apply in lieu of the rates and charges in Section 4.7.45 of this Tariff.

If the Customer is moved to Business Long Distance 50 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 50 Connections 2 Plus Service 2 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

At the end of the 2-year term, the Customer will be moved to Business Long Distance 50 Connections 2 Service for an additional 2-year term, as specified in Section 3.7.22 of this tariff, unless otherwise specified.

3.7 Custom Business Services (continued)	
------------------------------------------	--

- 3.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year
 - (A) Business Long Distance 100 Connections 2 Plus Service 2 Year is a custom combination switched TFS, outbound, and calling card Flat Rate Service available to Business Customers that:
 - .1 request to be provisioned under this optional Service;
 - .2 utilize Switched Access to reach the long distance network for | outbound calling and/or utilize Switched Access to receive calls from | the long distance network for TFS; and N

- 3.7 Custom Business Services (continued)
 - 3.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year
 - (A) (continued)
 - .3 subscribe to and maintain the following products or services provided by a SBC affiliate: (1) one DSL service or shared web hosting or dedicated Internet access or SBC® Yahoo!® dial-up Internet access or T1 Integrated Access or dedicated web hosting; and (2) a minimum of two Centrex station lines from a SBC Affiliate, or, a minimum of one business access line under a term agreement from a SBC Affiliate, or, a Business SolutionsSM, CompleteLink®, SimpleLink, Plexar I®, Centrex with Classic Feature PackageSM, Pacific Bell Instant OfficeSM:, Power Office, The Business Plan, Custom Biz SaverSM, Local Usage SaverSM or Ameritech Centrex Service (ACS) service package as defined in Section 1 of this Tariff or, subscribe to all of the following from a SBC Affiliate:
 - .a a minimum of one business access line, and,
 - .b an inside wire maintenance product associated with each business access line, and,
 - .c at least one instance of Caller ID, and,
 - .d at least one instance of each of any three call control features from Group D Package as defined in Section 1 of this Tariff from a SBC Affiliate.

N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year
 - (A) (continued)
 - .4 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Customer also subscribes to the products or services described in Section 3.7.46 (A).3 of this Tariff;
 - .5 associate the billing for the products or services of a SBC Affiliate that are described in Section 3.7.46 (A).3 of this Tariff with the Customer's BTN for Business Long Distance 100 Connections 2 Plus Service 2 Year. Exceptions to this requirement are: (a) Cingular service and (b) DSL service when subscription to an additional BTN was required for DSL provisioning by the SBC Affiliated LEC or Affiliated CLEC for technical reasons. The qualifying DSL service billed under a second BTN must be physically located at the same business premises as the local service to whose BTN the Business Long Distance 100 Connections 2 Plus Service 2 Year will be billed; and
 - .6 commit to an MMC of \$100 per month for a 2-year term plan and sign a written term plan agreement with the Company.

For rules and regulations regarding the MMC and term plans, see Section 2.26 of this Tariff. This Service is established at the BTN level and is only available for a single BTN. Service is only available for Customers that commit to a 2-year term plan.

Ν

- 3.7 Custom Business Services (continued)
 - 3.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year (continued)
 - (B) The Customer may subscribe to Business Long Distance 100 Connections 2 Plus Service 2 Year for outbound Service only, TFS only or for both outbound and TFS for a single BTN. The start of Service date may be on or after the installation date of the required products or services described in Section 3.7.46 (A).3 of this Tariff.
 - (C) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (D) For TFS, Customers may subscribe to TFS with or without the ability to receive intrastate Service from the Company.

- 3.7 Custom Business Services (continued)
 - 3.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year (continued)
 - (E) The Customer's usage rate for each call is based on a 2-year term plan.
 - (F) Outbound and TFS calls and calls billed to the Calling Card Option 2 are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.46 of this Tariff.
 - (G) If the Customer fails to maintain the required products or services described in Section 3.7.46 (A).3 of this Tariff, the Customer will no longer qualify for Business Long Distance 100 Connections 2 Plus Service 2 Year and will be moved to Business Long Distance 100 for an additional 2-year term, unless the Customer selects an alternative Service. If the Customer is moved to Business Long Distance 100, the rates and charges in Section 4.7.13 of the Tariff will apply in lieu of the rates and charges in Section 4.7.46 of this Tariff.

If the Customer is moved to Business Long Distance 100 or any alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment for Business Long Distance 100 Connections 2 Plus Service 2 Year, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff. At the end of the 2-year term, the Customer will be moved to Business Long Distance 100 Connections 2 for an additional 2-year term, as specified in Section 3.7.23 of this tariff, unless otherwise specified by the Customer.

SECTION 3 -	DESCRIPTION	OF	SERVICES
000110110	DECOIDE HOLD	<u> </u>	

3.7 Custom Business Services (continued)

- 3.7.47 Value Plans
 - (A) The Value Plans are custom combination switched TFS, outbound, and calling card Flat Rate optional pricing plans. Unless otherwise specified in the description of the rate option described in Section 3.7.47 (H) of this Tariff, the Value Plans are available to new and existing Business Customers that:
 - .1 request to be provisioned under this optional pricing plan;
 - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
 - .3 commit to the MMC as specified in the description of the rate option selected by the Customers;
 - .4 subscribe to and maintain at least one business access line of an SBC Affiliate; and
 - .5 commit to a 1-year term plan or commit to 2-year term plan and sign a written term plan agreement with the Company. For rules and regulations regarding term plans, see Section 2.26 of this Tariff.

N

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.47 Value Plans (continued)
 - (B) The Value Plans are established at the BTN level and are only available for a single BTN.
 - (C) The Customer may subscribe to any of the Value Plans for outbound Service only, TFS only or for both outbound and TFS for a single BTN. Business Customers subscribing to the Value Plans may also subscribe to the Calling Card - Option 2 at the rates described in Section 4.7.47 of this Tariff.
 - (D) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (E) The Customer's usage rate for each call is based on the rate option selected by the Customer and whether the Customer subscribes to Service on a 1year or 2-year term plan.
 - (F) A description of the billing increments applicable to the rate option selected by the Customer may be found in Section 3.7.47 (H) of this Tariff.

Ν

3.7

SECTION 3 - DESCRIPTION OF SERVICES

Custom Bus	siness Services (continued)	N
3.7.47 Val	ue Plans (continued)	
(G)	Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this plan. If the Customer is moved to an alternative Service and the Customer's MMC and term plan commitment is equal to or greater than the MMC and term plan commitment under one of the Value Plan rate options, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.	 N

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7	Custom	Business	Services	(continued)	
-----	--------	----------	----------	-------------	--

- 3.7.47 Value Plans (continued)
 - (H) Rate Options
 - .1 Business Domestic Value Saver 15

Business Domestic Value Saver 15 is available to Customers that commit to (1) an MMC of \$15 per month for a 1-year term plan or (2) an MMC of \$15 per month for a 2-year term plan and sign a written term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Domestic Saver 15 unless an alternative plan is selected by the Customer. Calls are billed in increments of six (6) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (A) of this Tariff.

Ν

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)	
------------------------------------------	--

- 3.7.47 Value Plans (continued)
 - (H) Rate Options (continued)
 - .2 Business Long Distance Value 50

Business Long Distance Value 50 is available to Customers that commit to (1) an MMC of \$50 per month for a 1-year term plan or (2) an MMC of \$50 per month for a 2-year term plan and sign a written term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Long Distance 50 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (B) of this Tariff.

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Ci	ustom Bus	siness Serv	vices (con	tinued)
--------	-----------	-------------	------------	---------

3.7.47 Value Plans (continued)

- (H) Rate Options (continued)
 - .3 Business Long Distance Value 100

Business Long Distance Value 100 is available to Customers that commit to (1) an MMC of \$100 per month for a 1-year term plan or (2) an MMC of \$100 per month for a 2-year term plan and sign a written term plan agreement with the Company. Customers who cancel or discontinue a business access line of a SBC Affiliate or whose service is refused, cancelled or discontinued by an SBC Affiliate shall forfeit eligibility for rates under this rate option and will be moved to Business Long Distance 100 unless an alternative plan is selected by the Customer. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. For rates and charges, see Section 4.7.47 (C) of this Tariff.

Ν

Ν

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.48 Business Unlimited Long Distance Plans
 - (A) Business Unlimited Long Distance Plans are custom combination switched TFS, outbound, and calling card optional pricing plans. For a monthly recurring charge, the Customer receives unlimited intrastate and interstate one plus (1+) Direct-Dialed MOU. Switched TFS calls and calls billed to a calling card are billed on a usage sensitive basis. For rates and charges, see Section 4.7.48 of this Tariff.
 - (B) Business Unlimited Long Distance Plans are available to new and existing Business Customers that:
 - .1 request to be provisioned under this optional pricing plan;
 - .2 utilize Switched Access to reach the long distance network for outbound calling and/or utilize Switched Access to receive calls from the long distance network for TFS;
 - .3 subscribe to and maintain or currently subscribe to and maintain at least one but not more than five business access lines of an SBC Affiliate that are associated with the qualifying BTN;

Ν

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.48 Business Unlimited Long Distance Plans (continued)
 - (B) (continued)
 - .4 subscribe to and maintain or currently subscribe to and maintain: a business access line of a SBC Affiliate plus any feature package of a SBC Affiliate listed below or a business access line with term of a SBC Affiliate plus any feature package of a SBC Affiliate listed below: (a) SimpleLink[™] or (b) Pacific Bell Instant Office[™] or Business Essentials[™] or (c) Business Solutions[™] or (d) one Measured Rate Service plus The Works® or (e) one Measured Rate Service plus PowerPack or (f) one Measured Rate Service plus Business Preferred[™] or (g) Power Office[™] or (h) Centrex Service (1 to 5 stations lines only) or (i) multiline plus Caller ID and series hunting or (i) Caller ID and any two of the following features on a single business line: Call Waiting, Call Waiting ID, Call Forwarding, Busy Call Forwarding, Busy Call Forwarding-Extended, Delayed Call Forwarding, Select Call Forwarding, Remote Access Call Forwarding, Repeat Dialing[™], Call Screen, Speed Calling 8[™], 30 Member Speed Calling, Call Return, Three Way Calling, Priority Ringing[™], PRIVACY MANAGER®, Call Transfer Disconnect, Auto RedialTM or Priority Call; and
 - .5 Reserved for future use

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.48 Business Unlimited Long Distance Plans (continued)
 - (B) (continued)
 - .6 commit to a 1-year term plan for the Business Unlimited Long Distance Plan. If the Customer discontinues Service prior to the expiration of the Business Unlimited Long Distance term plan agreement, the early termination fee applies pursuant to Section 2.26 of this Tariff. The Customer may upgrade, or downgrade, to another Business Unlimited Long Distance Plan; and the Company will waive the early termination fee associated with the change in plans if the Customer commits to another 1-year term and the number of business access lines match the Business Unlimited Long Distance Plan described in Section 3.7.48 (I) of this Tariff.
 - (C) Except for Customers subscribing to one of the Company's High Volume Calling Plans, the Company will waive the early termination fee for Business Customers cancelling their existing term plan agreement and committing to a 1-Year term plan for Business Unlimited Long Distance Plans.
 - (D) Business Unlimited Long Distance Plans are only available for a single BTN.

N

N

Ν

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.48 Business Unlimited Long Distance Plans (continued)
 - (E) The Customer may subscribe to Business Unlimited Long Distance Plans for outbound Service only or for both outbound and TFS for a single BTN. Business Customers subscribing to the Business Unlimited Long Distance Plans may also subscribe to the Calling Card - Option 2 at the rates described in Section 4.7.48 of this Tariff.
 - (F) Customers may subscribe to the Business Unlimited Long Distance Plans for the provision of interstate and intrastate InterLATA service; interstate, intrastate InterLATA, and intrastate IntraLATA service; or intrastate IntraLATA service only.
 - (G) Toll free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3.6 of this Tariff for optional features, rules and regulations, and general information regarding TFS.
 - (H) For switched TFS calls and calls billed to the Calling Card Option 2, calls are billed in increments of six (6) second subject to a minimum connect time (initial period) of thirty (30) seconds.

SECTION 3 - DESCRIPTION OF SERVICES

- 3.7 Custom Business Services (continued)
 - 3.7.48 Business Unlimited Long Distance Plans (continued)
 - (I) Customers with more than five (5) business access lines are not eligible for this plan. All business access lines under a participating BTN must be provisioned on this plan.

Customers who cancel or discontinue any of the qualifying products, services or features or whose Service is refused, cancelled or discontinued by the Company or those companies listed in Section 3.4.48 (B) .4 or .5 shall forfeit eligibility for rates under this Service. Customers continuing to presubscribe to the Company will be moved to Long Distance for Business.

(J) Every WTN within the participating BTN must be provisioned on this Service. Customers found in violation shall forfeit eligibility for rates under this plan and will be moved to Long Distance for Business unless an alternative plan is selected by the Customer. Early termination charges may apply as described in Section 2.26 of this Tariff. If the Customer is moved to an alternative Service and the Customer's term plan commitment is equal to or greater than the term plan commitment under Business Unlimited Long Distance Plans, the Company will credit the Customer's account for the amount of any early termination charges as described in Section 2.26 of this Tariff.

N

SECTION 3 - DESCRIPTION OF SERVICES

3.7 Custom Business Services (continued)

3.7.48 Business Unlimited Long Distance Plans (continued)

(K) Certain restrictions apply. Business Unlimited Long Distance Plans may not be used in conjunction with the following: auto dialers, long distance Internet access, call center applications including but not limited to autodialers, PBX trunks, ground start line or trunks, ISDN services, foreign exchange services, public telephone services, public access smart-pay phones, analog to digital conversion digital PBX services, WATS services, PBX/PABX/EABX services, non-square electronic key telephone systems, hybrid key telephone systems, predictive calling/dialing systems, automatic outbound dialing systems, any type of automatic call distribution system, or the functional equivalent of any such system listed above. If the Company determines that the Customer is in violation of above listed restrictions, the Customer shall forfeit eligibility for rates under this plan and will be moved to Long Distance for Business unless an alternative plan is selected by the Customer.

Ν

4.7 Custom Business Services (continued)

4.7.18 Business Domestic Saver 15 Connections 3 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30 Seconds	Each Add'l 1 Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ν

Т

4.7 Custom Business Services (continued)

4.7.19 Business Long Distance 50 Connections 3 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ń

N

4.7 Custom Business Services (continued)

4.7.20 Business Long Distance 100 Connections 3 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30 Seconds	Each Add'l 1 Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ν

Т

Т

Ν

Ν

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.21 Business Domestic Saver 15 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Т

Ν

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.22 Business Long Distance 50 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7 Custom Business Services (continued)

4.7.23 Business Long Distance 100 Connections 2 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ν

Т

Ň

4.7 Custom Business Services (continued)

4.7.24 Business Domestic Saver 15 Connections 1 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30 Seconds	Each Add'l 1 Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Т

4.7 Custom Business Services (continued)

4.7.25 Business Long Distance 50 Connections 1 Service

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ŋ

Т

4.7 Custom Business Services (continued)

4.7.26 Business Long Distance 100 Connections 1 Service (continued)

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Т

N

N

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued	4.7	Custom	Business	Services	(continued
-----------------------------------------	-----	--------	-----------------	----------	------------

4.7.33 Reserved for future use

4.7.34 Reserved for future use

Issued: July 16, 2003

Ň

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7.38 Business Domestic Saver 15 Connections 1 Plus Service 2 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7.39 Business Long Distance 50 Connections 1 Plus Service 2 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7 Custom Business Services (continued) Ň 4.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and 0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card -Option 2 is 0.07 for the initial thirty (30) seconds and 0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff. 4.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year The usage rate for outbound calls and TFS calls is 0.045 for the initial thirty (30) seconds and 0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card -Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff. 4.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year The usage rate for outbound calls and TFS calls is 0.045 for the initial thirty (30) seconds and 0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card -Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff. N

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

4.7.45 Business Long Distance 50 Connections 2 Plus Service 2 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30) seconds and \$0.0015 for each additional one (1) second. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year

The usage rate for outbound calls and TFS calls is \$0.045 for the initial thirty (30)seconds and \$0.0015 for each additional one (1) second. The usage rate for fullyautomated, operator assisted, and operator dialed calls billed to the Calling Card -Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for eachadditional one (1) second. The per call charge may be found in Section 4.1.1(B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

Ν

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.47 Value Plans

(A) Business Domestic Value Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer	Initial 30	Each Add'l
Commitment	Seconds	6 Seconds
1 Year Term Plan	\$0.0450	\$0.009
2 Year Term Plan	\$0.0450	\$0.009

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.015 for each additional six (6) seconds. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

N

Ν

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.47 Value Plans

(B) Business Long Distance Value 50

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.075 for the initial thirty (30) seconds and \$0.0025 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

N

SECTION 4 - RATES AND CHARGES

4.7 Custom Business Services (continued)

4.7.47 Value Plans

(C) Business Long Distance Value 100

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	Initial 30	Each Add'l 1
	Seconds	Second
1 Year Term Plan	\$0.0450	\$0.0015
2 Year Term Plan	\$0.0450	\$0.0015

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.0024 for each additional one (1) second. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

- 4.7 Custom Business Services (continued)
 - 4.7.48 Business Unlimited Long Distance Plans

Business Unlimited Long Distance Plans

The per minute usage rate for switched TFS is as follows:

		Each Add'l 6 Seconds
Switched TFS	\$0.0450	\$0.0090

The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 is \$0.07 for the initial thirty (30) seconds and \$0.014 for each additional six (6) seconds. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (B), and Section 4.1.2 (C) of this Tariff.

The MRC for unlimited interstate and intrastate 1+ outbound calling is as follows:

Number of Access Line Subscribed to Business Unlimited Long Distance Plans	MRC
1	\$20
2	\$40
3	\$60
4	\$80
5	\$100

N